<%namespace name="tools" file="tools.tex"/>
<%! import os %>

The experiments presented in the previous sections \citep{VanDeGraaff1976,Vellinga1981}, resulted in a final theory on scale relations and the translation of small scale results to prototype situations \citep{Vellinga1986}. In order to verify this theory, large scale model tests are performed in 1980 and 1981 in the Deltaflume of Delft Hydraulics, currently known as Deltares \citep{Vellinga1984}. The Deltaflume is approximately 230m long, 5m wide and 7 to 9m deep.

Five experiments are performed, as presented in \autoref{tab:experiments_M1263III}. Tests 1, 2 and 5 had a constant surge level, while tests 3 and 4 had a variable surge level with a course depicted in \autoref{fig:hydraulicconditions_Test3} and \autoref{fig:hydraulicconditions_Test4} respectively.

\begin{table}[H]
  \caption{Overview of experiments}
  \label{tab:experiments_M1263III}
  \begin{center}
    \begin{tabular}{lllllll}
      \hline
      Experiment & Scale & Profile     & Sediment & Water & Wave   & Wave   \\ 
                 &       & contraction & diameter & depth & height & period \\ 
      \hline
      1          & 5     & 1.91        & 225      & 4.2   & 1.50   & 5.4    \\ 
      2          & 5     & 1.27        & 225      & 4.2   & 1.50   & 5.4    \\ 
      3          & 5     & 1.27        & 225      & 4.2   & 1.50   & 5.4    \\ 
      \hline
      4          & 3.27  & 1.91        & 225      & 4.2   & 1.85   & 5.0    \\ 
      5          & 1     & 1           & 225      & 5.0   & 2.00   & 7.6    \\ 
      \hline
    \end{tabular}
  \end{center}
\end{table}

${tools.image(os.path.join(input_path, 'Deltaflume_M1263/III/Test-3/'), 'hydraulicconditions_Test3.eps', 'Boundary conditions for test 3')}
${tools.image(os.path.join(input_path, 'Deltaflume_M1263/III/Test-4/'), 'hydraulicconditions_Test4.eps', 'Boundary conditions for test 4')}

As in the previous sections, the profile developments are compared to the measurements in \autoref{fig:deltaflumem1263_iii_t1_profile} to \autoref{fig:deltaflumem1263_iii_t5_profile}. Moreover, erosion patterns and volumes and retreat distances are analyzed in \autoref{fig:deltaflumem1263_iii_t1_erovol} to \autoref{fig:deltaflumem1263_iii_t5_erovol}. \autoref{fig:Deltaflume_M1263_III_bss} provides an overview of the Brier Skill Scores of the profile development for the different tests.

${tools.image('Deltaflume_M1263_III/Test-1/', 'final_profiles.eps', 'Profile development during test 1', 'deltaflumem1263_iii_t1_profile')}
${tools.image('Deltaflume_M1263_III/Test-2/', 'final_profiles.eps', 'Profile development during test 2', 'deltaflumem1263_iii_t2_profile')}
${tools.image('Deltaflume_M1263_III/Test-3/', 'final_profiles.eps', 'Profile development during test 3', 'deltaflumem1263_iii_t3_profile')}
${tools.image('Deltaflume_M1263_III/Test-4/', 'final_profiles.eps', 'Profile development during test 4', 'deltaflumem1263_iii_t4_profile')}
${tools.image('Deltaflume_M1263_III/Test-5/', 'final_profiles.eps', 'Profile development during test 5', 'deltaflumem1263_iii_t5_profile')}

${tools.image('Deltaflume_M1263_III/', 'bss.eps', 'Overview of Brier Skill Scores')}

${tools.image('Deltaflume_M1263_III/Test-1/', 'fig2.eps', 'Development of erosion volumes during test 1', 'deltaflumem1263_iii_t1_erovol')}
${tools.image('Deltaflume_M1263_III/Test-2/', 'fig2.eps', 'Development of erosion volumes during test 2', 'deltaflumem1263_iii_t2_erovol')}
${tools.image('Deltaflume_M1263_III/Test-3/', 'fig2.eps', 'Development of erosion volumes during test 3', 'deltaflumem1263_iii_t3_erovol')}
${tools.image('Deltaflume_M1263_III/Test-4/', 'fig2.eps', 'Development of erosion volumes during test 4', 'deltaflumem1263_iii_t4_erovol')}
${tools.image('Deltaflume_M1263_III/Test-5/', 'fig2.eps', 'Development of erosion volumes during test 5', 'deltaflumem1263_iii_t5_erovol')}